Author's response to reviews

Title: Thyroid shields and neck exposures in cephalometric radiography.

Authors:

Philippe P Hujoel (hujoel@u.washington.edu)
Lars Hollender (larsholl@u.washington.edu)
Anne-Marie Bollen (mine@u.washington.edu)
John Young (ydy2@u.washington.edu)
Joana Cunha-Cruz (silvajcc@u.washington.edu)
Molly McGee (mkmcgee@u.washington.edu)
Alex Grosso (grosso@u.washington.edu)

Version: 6 Date: 12 June 2006

Author's response to reviews:

We have changed the competing interest section according to the editor's suggestion.